Author's response to reviews

Title: Is it possible to identify cases of coronary artery bypass graft postoperative surgical site infection accurately from claim data? A multi-model comparison study over 2005-2008

Authors:

Tsung-Hsien Yu (ericthyu@ntu.edu.tw)
Yu-Chang Hou (houyc0115@mail2000.com.tw)
Kuan-Chia Lin (kuanchia@ntunhs.edu.tw)
Kuo-Piao Chung (kpchung@ntu.edu.tw)

Version: 5  Date: 4 May 2014

Author's response to reviews: see over
Dear Associate Editor, Dr. van Mourik,

Thank you for your helpful comments and encouragement, which have helped us tremendously in revising the manuscript. In this revision, we made two modifications (see italicised and bold sentences with underlining) according to Dr. van Mourik’s suggestions. And we also sent this manuscript for language editing.

Yours sincerely,

Kuo-Piao Chung, PhD
Professor, Institute of Health Policy and Management
College of Public Health, National Taiwan University